302' Copyright Ron Burback 1997IDL and the ACFRon BurbackLecture 17

303' Copyright Ron Burback 1997IDL and the ACFnInterface Definition Language—Defines the interfacenAttribute Configuration File—Defines the options

304' Copyright Ron Burback 1997Simple IDL typesnfloatndoublencharnbooleannbytenvoidnhandle_tnintnsmallnshortnlong n hyper

n string n arrays n ref n pipe n error_status_t

305' Copyright Ron Burback 1997Pointersnref, reference pointers, limited usage across the rpc, can not change valuenptr, full pointer

306' Copyright Ron Burback 1997ArraysnFixed—fixed size—all sentnConformant—variable size—all sentnVariable—variable size—only portion sent

307' Copyright Ron Burback 1997Structured IDL typesnconstntypedefnstructnenumnunion

308' Copyright Ron Burback 1997Context handlesncontext_hdl_tnallows for a server and client to save context between calls

309' Copyright Ron Burback 1997Source of ArgumentsnInnOutnIn, Out

310' Copyright Ron Burback 1997Type of callnidempotentnbroadcastnmaybe

311' Copyright Ron Burback 1997Port SelectionTCP, UDPnstatic port number

312' Copyright Ron Burback 1997Four types of Bindingsnstring - by pass the cds and go directly to this locationnauto - binding handle done in stub code, no programmer visibilitynimplicit - binding handle managed by the user, one global variablenexplicit - binding handle managed by the user, one for each rpc call

313' Copyright Ron Burback 1997In the AFCncode/nocode: generation of stub codenin_line/out_of_line: is the marshalling code in line or not?nheap: parameters are allocated in the heap in the server. By default use the stack.nrepresent as/transmit as: type casting

314' Copyright Ron Burback 1997Exceptions across the rpcncomm_statusnfault_status